Title: Inadequate glucose control in type 2 diabetes is associated with impaired lung function and systemic inflammation: a cross-sectional study

Authors:

Rodolfo J Dennis (rdennis@javeriana.edu.co)
Dario Maldonado (dmalдонado@cable.net.co)
Maria X Rojas (mxrojas@javeriana.edu.co)
Pablo Aschner (paschner@cable.net.co)
Martin Rondón (martin.rondon@javeriana.edu.co)
Laura Charry (lcharry@javeriana.edu.co)
Alejandro Casas (acasas@neumologica.org)

Version: 4 Date: 23 July 2010

Author's response to reviews:

Title: Inadequate glucose control in type 2 diabetes is associated with impaired lung function and systemic inflammation: a cross-sectional study

Journal: BMC Pulmonary Medicine

MS ID: 1314729701316764

Dear Editors,

We have now done the following changes to the format of the manuscript, so that the published article conforms to journal style:

1) Author list: the list of authors in the manuscript is now written exactly as they are in the submission system, both in style and order.

2) Key words were removed.

3) Tables- the visible lines were removed.

We have also checked the manuscript for typographical errors.

Sincerely,

Rodolfo J Dennis - on behalf of all authors.